Author’s response to reviews

Title: Impact of Y181C and/or H221Y Mutation Patterns of HIV-1 Reverse Transcriptase on Phenotypic Resistance to Available Non-Nucleoside and Nucleoside Inhibitors in China

Authors:

Wei Guo (guowei0922@126.com)
Hanping Li (hanningline@163.com)
Daomin Zhuang (zdm0705@hotmail.com)
Liyan Jiao (ysmx_2000@sina.com)
Siyang Liu (aliusy_0@sina.com)
Lin Li (dearwood@sina.com)
Yongjian Liu (yongjian325@sina.com)
Tao Gui (guibaodj@126.com)
Lei Jia (jjialeitsinghua@163.com)
Jingyun Li (lijy@nic.bmi.ac.cn)

Version: 6
Date: 23 April 2014

Author’s response to reviews: see over
Dear Editor:

Thank you for your email.

We have made language corrections according to the reference.

Thank you again for your Email. I look forward to hearing from you.

Sincerely,

Jingyun Li

Department of AIDS Research
State Key Laboratory of Pathogen and Biosecurity
Beijing Institute of Microbiology and Epidemiology
20 Dongda Street
Fengtai District
Beijing 100071
China
Phone: 86-10-66948566
Fax: 86-10-63842689
E-mail: lijy@nic.bmi.ac.cn